RESEARCH

Open Access

Adult cognitive outcomes in phenylketonuria: explaining causes of variability beyond average Phe levels



Cristina Romani^{1*}, Filippo Manti², Francesca Nardecchia², Federica Valentini³, Nicoletta Fallarino³, Claudia Carducci⁴, Sabrina De Leo⁵, Anita MacDonald⁶, Liana Palermo⁷ and Vincenzo Leuzzi²

Abstract

Objective: The objective was to deepen the understanding of the causes of individual variability in phenylketonuria (PKU) by investigating which metabolic variables are most important for predicting cognitive outcomes (Phe average vs Phe variation) and by assessing the risk of cognitive impairment associated with adopting a more relaxed approach to the diet than is currently recommended.

Method: We analysed associations between metabolic and cognitive measures in a mixed sample of English and Italian early-treated adults with PKU (N = 56). Metabolic measures were collected through childhood, adolescence and adulthood; cognitive measures were collected in adulthood. Metabolic measures included average Phe levels (average of median values for each year in a given period) and average Phe variations (average yearly standard deviations). Cognition was measured with IQ and a battery of cognitive tasks.

Results: Phe variation was as important, if not more important, than Phe average in predicting adult outcomes and contributed independently. Phe variation was particularly detrimental in childhood. Together, childhood Phe variation and adult Phe average predicted around 40% of the variation in cognitive scores. Poor cognitive scores (> 1 SD from controls) occurred almost exclusively in individuals with poor metabolic control and the risk of poor scores was about 30% higher in individuals with Phe values exceeding recommended thresholds.

Conclusions: Our results provide support for current European guidelines (average Phe value = $< 360 \mu mol/l$ in childhood; = $< 600 \mu mo/l$ from 12 years onwards), but they suggest an additional recommendation to maintain stable levels (possibly Phe SD = $< 180 \mu mol/l$ throughout life).

Public significance statements: We investigated the relationship between how well people with phenylketonuria control blood Phe throughout their life and their ability to carry out cognitive tasks in adulthood. We found that avoiding blood Phe peaks was as important if not more important that maintaining average low Phe levels. This was particularly essential in childhood. We also found that blood Phe levels above recommended European guidelines was associated with around 30% increase in the risk of poor cognitive outcomes.

Keywords: PKU cognitive outcomes, Phe associations, Phe fluctuations, Phe variations, PKU guidelines

* Correspondence: C.Romani@Aston.ac.uk

¹School of Life and Health Sciences, Aston University, Aston Triangle, Birmingham, England B4 7ET, UK

Full list of author information is available at the end of the article



© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

Phenylketonuria (PKU; OMIM#261600) is an inherited metabolic disease where a genetic error results in a partial or complete de-activation of the enzyme phenylalanine hydroxylase (PAH) which normally metabolizes the amino acid phenylalanine (Phe; E.C. 1.14.16.1) into tyrosine (a precursor of dopamine). Phe accumulation results in several and still incompletely known negative effects on the postnatal development of the brain as well as on the functioning of the mature brain [1]. Fortunately, these negative consequences can be controlled by adopting, since birth, a Phe-restricted diet and protein supplementation. There is no question that a low Phe diet must be followed throughout childhood to achieve good cognitive outcomes [2]. However, several questions remain open [3]. We need to know more about: 1. which measures are most important to consider for dietary control (Phe average vs Phe fluctuations); 2. the impact of dietary control on different cognitive functions and possible interactions with age; and 3. which Phe value should be considered safe at different developmental age; there is uncertainty especially regarding the levels which are safe after early childhood. The purpose of this study is to provide some evidence relevant to these questions by analysing the performance of a mixed group of English and Italian early treated adults with PKU (from now on AwPKU) in relation to the current and historical blood phenylalanine control.

Which metabolic measure? (average Phe levels vs Phe variation)

Blood Phe levels are usually measured with the assumption that they correlate with levels in the brain (see Leuzzi et al. [4]; Pietz et al. [5]; Rupp et al. [6], but also Brumm et al. [7], Moats et al. [8]; Schindeler et al. [9] for no relationship. Different measures of blood Phe have been found to correlate with cognitive performance, but their relative contribution is unclear (from now on Phe without qualification refers to blood Phe).

Most studies have assessed the impact of dietary control by considering either *current Phe levels* or *average levels* across a time period (also referred to as IDCindex of dietary control). Average levels have generally been calculated as a *mean* of *yearly median* values or, more rarely as a *mean* of *half-year median* values (for examples of this latter measure see Pietz et al. [10]; Vilaseca et al. [11]). These studies have shown that current Phe levels as well as average Phe levels are good predictors of cognition (for examples of positive associations in adults across cognitive functions see Brumm et al. [7]; Romani et al. [12]; for effects on IQ see Manti et al. [13]; Weglage et al. [14]; for effects on IQ in children, see Waisbren et al. [2]). Note, however, that effects are limited when only a restricted set of tasks is used [15, 16] and/or when only current Phe level has been considered; for example, effects of *current* Phe on IQ have been inconsistent across studies (see Jahja et al. [17]; Moyle et al. [18] for positive and/or marginal results; see Koch et al. [19]; Feldmann et al. [20]; Pietz et al. [10], for no correlation).

Phe variation (also referred to as Phe fluctuation by some authors) has also been shown to predict cognition. Phe variation has generally been measured as a *mean* of *yearly SD* of Phe values [21–23]. Most studies have considered children and found that indexes of variation predict IQ (Burgard et al. [24]; Hood et al. [25]; marginally significant results in Anastasoaie et al. [21]; see also Vilaseca et al. [11] for results with a mixed-age group), executive functions [22, 24], motor control [26], white matter integrity [27]; for a review across functions, see Cleary et al. [28]. There is more limited evidence that Phe variation predicts cognitive outcomes long-term, since studies on adult patients are lacking.

Viau et al. [23] studied a mixed sample of children and young adults (N = 55) and assessed the impact of current and historical Phe on cognition. They reported limited correlations with Phe averages and no correlations at all with Phe SD. However, cognition was measured only with limited subtests from the WAIS and the WISC (Block design, Symbol Search and Verbal IQ or Verbal comprehension). Our previous study on a sample of 37 English AwPKU, early treated and with good metabolic control, showed significant effects of both historical Phe average and Phe SD (0-10, 11-16, 17+) on adult cognitive performance measured through IQ and an ad-hoc PKU battery of cognitive tasks [12].¹ Importantly, however, these results did not provide information on the relative contribution of Phe average and Phe SD to cognitive outcomes. These two measures are, in principle, independent of one another. Two individuals can maintain the same average Phe level, but one may show little variation around the mean, with values very similar to one another, while another may show a lot of variation. Thus, both average Phe and Phe variation may contribute independently to good cognitive outcomes. However, in practice, these two measures are highly correlated in PKU populations, because individuals who maintain a lower Phe average, also maintain a more consistent low Phe diet [11, 12, 23, 25].

Hood et al. [25] reported some independent contributions of Phe SD, but they only assessed relationships in children and with limited cognitive measures (they found an independent contribution of childhood SD 5–

¹Life-time Phe SD predicted FIQ and visuo-motor coordination; <u>childhood Phe SD</u> predicted visuo-spatial attention, complex EF and memory and learning; <u>adolescence Phe SD</u> predicted spoken language and <u>adult Phe SD</u> predicted sustained attention.

10 at years or after 10 years on matrix reasoning and number of non-responses in an N-back task). In our study, we aim to assess an independent contribution of Phe SD on adult cognitive outcomes assessed more comprehensively.

Individual variation in cognitive outcomes

While it is clear that cognitive outcomes depend on metabolic control, the extent of this dependence is debatable.

One question relates to whether all effects of having PKU can be eliminated through dietary control [1]. We know that most early-treated AwPKU perform within the norm, but that, as a group, their performance is worse than the controls. What we do not know, however, is whether the whole distribution of cognitive scores is shifted so that even performance at the highend of the distribution is affected, or, instead, it is only the lower end of the distribution which is affected, where individuals are likely to have maintained poor dietary control. The first option will indicate that there are some fixed costs of having PKU which are not avoidable even maintaining a low Phe diet following current treatment guidelines. The second option instead, will indicate that a strict diet can completely eliminate the cognitive impact of having PKU.

A second, related question concerns the safe target range for blood Phe control at different ages. Current European guidelines advise to maintain Phe average levels below 360 µmol/L, before 12 years of age and below 600 µmol/L thereafter [29, 30]. American guidelines are even more strict recommending 120–360 µmol/ L throughout life (American College of Medical Genetics and Genomics, ACMG) [31]. However, even the European guidelines have been criticized for being over stringent [32]. This is because there is little evidence of illeffects when guidelines are relaxed in adulthood [13] and even the evidence to advocate childhood Phe < 360 is not strong [33-36]. A way to examine this question is to examine the distributions of cognitive scores within the PKU group in relation to metabolic control (see Waisbren et al. [2] for analyses of children data). This will allow us to examine if there are discontinuities in the distributions of cognitive scores, with pathological scores starting to appear and/or become more frequent when a given metabolic value is exceeded and whether these boundaries are consistent with current guidelines. Additionally, the cost of not following guidelines can be quantified by comparing the rates of poor cognitive scores in individuals which have or have not followed guidelines.

A final, related question, is whether there are individuals who have maintained poor metabolic control, but still have escaped cognitive impact. This will show that there is variability on how negatively PKU affects cognition (see van Vliet et al. [37] for a review of extreme case).

In conclusion, our study has two related aims: 1. To compare the effects of protracted exposure of brain to Phe –best measured through average Phe levels— with the effects of Phe peaks –best measures through SD from the mean--, and possible interactions with age. We want to see whether *both* average Phe and Phe SD contribute to adult outcomes and whether these two measures have a different weight in childhood and ado-lescence/adulthood. 2. To assess cognitive variability in a population of adults with PKU to see a) whether effects are pervasive or limited to a portion of individuals, b) whether the Phe boundaries identified by current European guideline are meaningful and c) whether there are exceptional cases where good cognition is achieved in spite of poor metabolic control.

To achieve aims, we have combined results from English and Italian AwPKU tested with the same battery of tasks (N = 56). Italian and English sub-samples show similar patterns of cognitive impairments and relationships with current and historical Phe measures, justifying accruing results (Romani et al., unpublished data). The resulting sample is larger and more varied in terms of metabolic control than most sets reported in the literature allowing better assessment of correlations between metabolic and cognitive variables (current Phe range is 54–2081; SD = 403; compared, for example, to: Brumm et al. [7]: 157–1713; SD = 338; Channon et al. [38]: 221–1233; SD = 261; Jahjia et al. [17]: 66–1550; SD = 342; Smith et al. [39]: 200–1879).

Method

Recruitment

Fifty six early-treated adult PKU participants were tested: 19 Italian and 37 English. They were all diagnosed soon after birth as result of national newborn screening programs.

The 19 Italian AwPKU were recruited from the Clinical Centre for Neurometabolic Diseases Department of Human Neuroscience, Child Neurology and Psychiatry Unit, Sapienza University of Rome. Three participants were currently treated with Kuvan. Nineteen Italian control participants were recruited among friends and students of the researchers. They were matched to the Italian PKU participants for age and education. Among the Italian participants, 4 had a diagnostic Phe level > 600 μ mol/L but < 1200 μ mol/L; 15 participants had Phe > 1200 μ mol/L at birth.

The 37 English AwPKU participants were recruited from the Department of Inherited Metabolic Disorders at the University Hospitals Birmingham. They all had Phe > 1200μ mol/L at birth. The performance of this

sample on a larger set of tasks as been described in previous publications [12, 40, 41]. Thirty English healthy controls were recruited through an advertising volunteering website. They were matched to the English PKU participants for age and education.

All AwPKU treated in the English and Italian centres were invited to participate and were accepted in the study on a first come, first served basis. The English study received NHS ethical approval. The Italian study was approved by the local ethics committee. All participants provided informed consent to the study.

Metabolic measures

For both the English and the Italian PKU participants blood spots for blood Phe were taken regularly since diagnosis in early infancy and extensive records were available although there were limited data for a few participants (6 UK participants lacked or had very limited childhood data). We averaged Phe control in three age bands: childhood: 0-10 years old, adolescence: 11-16 years old, and adulthood: 17 years to present. We have also averaged measures throughout the life-time and considered current Phe level (for the Italian group, Phe has been measured immediately before the testing session/s or close to it; for the UK group, Phe has been measured immediately before the two testing sessions and averaged). We considered two types of measures: Phe average and Phe variation. Phe average in each band was calculated by taking the median values for each year and, then averaging the yearly values. The median is the value set halfway in a distribution of scores; it is generally used in the PKU literature rather than the mean because the median is not influenced by Phe variations. It is particularly, important to use the median in our study since we want to contrast a measure of central tendency (median, mean) with a measure of variation. Phe variation in each band was calculated by taking the SD for each year and then averaging yearly values in the band.

Cognitive assessment

Cognitive assessments were carried out in a quiet room at the clinical centres in Birmingham and Rome by one the psychologist on the team. The testing session for the Italian participants lasted between 2 and 3 h. The English participants were tested in two separate sessions of similar length (a less extensive set of tasks was administered to the Italian participants because of resource limitations). A few PKU participants were not able to attend the second testing session which resulted in some data points missing for some tests (N = 31 instead of 37).

IQ was measured using, the Wechsler Adult Intelligence Scale-Revised (WAIS-R, [42]) with the Italian participants and the Wechsler abbreviated scale of intelligence (WASI, [43]) with the English participants, which includes the following subtests: Vocabulary, Block Design, Similarities, and Matrix Reasoning. In addition, participants were given a set of tasks chosen from the larger set of tasks administered in our previous studies [12, 40]. We chose tests which either showed a strong difference between participants with PKU and controls and/or strong correlations with metabolic measures. We also gave precedence to tasks with non-linguistic stimuli which did not need adapting across languages. Therefore, we did not include tests of picture naming, reading, spelling and orthographic knowledge (spoonerisms, phoneme deletions). Accuracy in these tasks was very good and not related to metabolic measures [12]. Speed of processing was assessed with visual search tasks. To reduce the number of tasks tapping similar functions, we also did not administer the Tower of Hanoi, the lexical learning task, the Stroop, and nonword repetition. Measures of STM (digit span and Corsi span) and a baseline measure of peripheral speed of processing were included for completeness and because of mixed results from the literature (for impairments in digit span and nonword repetition see Palermo et al. [40]; for contrasting results see Brumm et al. [7], and Moyle et al. [18]; see also Jahja et al. [17], for deficits with increasing working memory load).

The following cognitive areas were assessed:

- 1. Visual Attention. This was assessed with four tasks [12, 40]: 1. Simple Detection: Press a response button as soon as a ladybird appears on the screen; 2. Detection with Distractors: Press a button when a ladybird appears on the screen alone or with a green bug; in the second part of the task the instruction was changed to press a button when a green bug appears on the screen alone or with a ladybird; 3. Feature Search: Detect a target among distractors not sharing features by pressing a ves no' button (e.g., a red ladybird among green or bugs); 4. Conjunction Search: Detect a target among distractors sharing features (e.g., red ladybird among red bugs and green bugs). Both reaction times (RT from now on) and accuracy measures (error rates) were taken.
- 2. *Visuo-motor Coordination*. This was assessed with two tasks: 1. *Grooved Pegboard Test* [44]: Put pegs into the holes of a board using only one hand as quickly as possible (short version with two trials one with the dominant and one with the non-dominant hand to match Italian and English samples) and 2. *Digit Symbol Task* [42]: Fill as many

boxes as possible with symbols corresponding to numbers (key with associations remains visible) in 90 s. *Trail Making Test A (TMT A)* [45, 46]: connect circles containing numbers in ascending order of the numbers as quickly as possible.

- 3. Complex Executive Functions. This was assessed with four tasks tapping skills such as planning, flexibility and abstract thinking: 1. The Wisconsin *Card Sorting Test (WCST)* 64 card version [47]: Discover the rules to match cards from a deck with four reference cards according to the shape, number or colour of the symbols on the card; feedback is provided to allow learning. Flexibility is required when the sorting rule is changed unknown to the participant and the new rule must be discovered. We used three different scores: total errors, number of perseverative responses and number of completed categories. 2. Difference in speed between Trail Making Test B-A (TMT B-A) [45, 46]. A involves connecting circles containing numbers in ascending order; B also involves connecting circles in ascending order, but alternating between circles containing numbers and letters. Only completion time is considered in this test; when, occasionally, an error is made, it is corrected by the examiner and this affects time to complete the task. 3 Fluency: For letter fluency: generate as many words as possible starting with a given letter in one minute of time (for Italian: P, F and L; Novelli et al. [48]; for English: C, F and L; Benton et al. [49]); for semantic fluency [50, 51]: generate as many names of animals as possible in one minute of time. This requires planning an efficient search through the lexicon.
- 4. Short-term Memory/Working Memory. This was assessed with two tasks: 1. *Digit Span*: Repeat a sequence of digits spoken by the examiner, soon after presentation; 2. *Corsi Block Tapping Test* [52]: The examiner taps a sequence of blocks and the participant must reproduce the sequence in the same order.
- 5. Sustained Attention This was assessed with the Rapid Visual Information Processing task (RVP; adapted from Sahakian et al. [53]): detect three target sequences of 3 digits by pressing the response key when the last number of the sequence appears on the screen. Scores are percentage correct.
- 6. Verbal Memory and Learning. This was assessed with *The Rey Auditory Verbal Learning Test* [54, 55] which asks for learning, immediate recall, and delayed recall of a list of 15 words. The list is presented five times and participants are asked to recall the words immediately after each presentation. After the 5th presentation (A5), an

interfering list (B1) is presented and participants are asked to recall this list and then, once again, the original list (A6) without a further presentation. Finally, participants are asked to recall the original list after a 20-min filled interval. Our scores include total number of errors across the five learning trials (A1–5); errors in recalling the words after an interfering list (A6); and, again, errors in delayed recall of the original list.

7. *Visual Memory and Learning.* This was assessed with the *Paired Associates Visual Learning* [56]: Learn to associate objects with locations.

Demographics and preliminary analyses Data analysis

For each participant, we computed z scores for each task using the relative (Italian or English) control group as reference. We also averaged z scores across tasks as a measure of overall cognitive performance. We report results of the PKU group using z-scores. Group differences of PKU from controls is examined through t-tests. Relationships between cognitive scores and Phe is examined with Pearson bivariate correlations. To reduce the number of variables per task, we did not carry out correlations with accuracy measures in search task (which are not impaired), and we only correlated for the TMT, the B-A condition; for the WCST, the total errors; and for the Rey, performance over 1–5 trials (learning) and in delay recall.

Participants

Table 1 shows demographic variables for age, gender, years of education and Phe control across age. Average Phe level increased across ages (diet became more relaxed), Phe variation remained more stable (see also Hood et a [25]., for similar results in children up to 18 years old).

Cognitive outcomes

Cognitive performance across tasks is shown in Table 2. Patterns of results are very similar to those reported previously with an overlapping sample of 37 AwPKU [40], except for the visual paired-associate learning which shows a modest group impairment. The tasks with the largest differences from controls were tasks of visual search measured in terms of speed of processing and task involving visuo-motor coordination (pegboard, digit symbol, TMT A). Executive functions in terms of flexibility and planning, (TMT B, verbal fluency²) and sustained attention were also impaired consistent with previous results (see for *speed of processing:* Albrecht

 $^{^2\}text{Note}$ these tasks have also a speed component, although not as strong as tasks measuring RTs.

	PKU Group		Control Group		
	N = 56		N = 49		
	Mean	SD	Mean	SD	
Age	26.8	6.4	26.5	6.3	
Education (in years)	14.3	1.9	14.7	1.8	
Gender (M/F)	21//35		21//28		
VIQ	101.1	12.9	110.8	11.4	
PIQ	102.7	15.2	110.1	11.1	
FIQ	102.2	14.4	112.3	11.3	
Childhood (0–10 years)					
Phe Average Median	457	213			
Phe Fluctuation	214	64			
Mean N obs. Per participant	201	141			
Adolescence (11–16 years)					
Phe Average Median	714	294			
Phe Fluctuation	162	56			
Mean N obs. Per participant	78	65			
Adulthood (17 years +)					
Phe Average Median	859	307			
Phe Variation	164	77			
Mean N obs. Per participant	63	67			
Lifetime					
Phe Average Median	655	263			
Phe Fluctuation (SD)	179	58			
Mean N obs. Per participant	341	213			
Current Phe	833	403			
Range	65–2081				

Table 1 Demographic and metabolic information for English and Italian PKU groups matched for age, gender and education, and for the whole group. Blood Phe measured in μ mol/L

et al. [57]; *visuo-motor coordination*: Griffiths et al. [58]; Pietz et al. [10]; *executive functions*: Smith et al. [39]; Brumm et al. [7]; *sustained attention*: Schmidt at al [55].; Bik-Multanowski et al. [59]; Weglage et al. [14]; Jahja et al. [17]).

Cognitive outcomes in relation to metabolic control

Table 3 shows bivariate Pearson r correlations between cognitive and metabolic measures. Correlations were extensive both for Phe average and Phe variations. Correlations were significant both with current and historical measures and for all tasks (except the Corsi span), although they were not systematic across all ages and types of metabolic measures. Significant correlations with lifetime measures (either average or SD) were found with IQ, speed in visual search, tasks tapping visuo-motor coordination, EF (WCST, TMT-B-A and

Table 2 Cognitive performance of the PKU group (English and Italian PKU participants; N = 56). Z scores calculated from respective control groups (N = 30 and N = 19). To facilitate interpretation, for all scores, higher Z-score reflect worse performance. Scores in bold are significantly higher than expected. ms. = milliseconds; sec. = seconds

	PKU Ov	verall 2	Z score
	Mean	SD	Diff from controls (t-test)
IQ	0.8	1.3	<i>p</i> < .001
Visual attention RTs			
Simple Detection - ms.	0.3	1	n.s.
Detention with Distractors – ms.	0.8	1.6	<i>p</i> < .01
Feature Search – ms.	1.5	2.3	<i>p</i> < .001
Conjunction Search m.	1.2	1.7	<i>p</i> < .001
Visual attention accuracy			
Detention with Distractors - % errors	0.2	1	n.s.
Feature Search - % errors	0.2	2.6	n.s
Conjunction Search - % errors	-0.1	0.8	n.s
Visuo-motor coordination			
Pegboard (sec.)	0.9	1.9	<i>p</i> < .01
Digit Symbol (%errors in 90 s.)	0.7	1.2	<i>p</i> < .01
Trail-Making Test A: (sec.)	0.8	1.9	<i>p</i> < .01
Executive functions			
WCST			
Total errors	0.4	1.6	n.s.
Perseverative responses	0.1	1.1	n.s.
N of Completed Categories	0.5	1.9	<i>p</i> = .06
Trail-Making Tests			
B (sec.)	0.6	1.5	<i>p</i> = .01
B-A (sec.)	0.5	1.7	p = .06
Verbal Fluency			
Letter (correct answers)	0.6	1	<i>p</i> < .01
Semantic (correct answers)	0.8	1.3	<i>p</i> < .001
Short term memory			
Digit span	0.3	1.1	n.s.
Corsi Block span	0.0	1.1	n.s.
Sustained attention			
RVP (% of errors)	0.7	1.3	p < .01
Learning			
Rey Auditory Verbal Learning Test			
Trial A1-A5 (% errors)	0.2	1.2	n.s.
Retention (% errors A6)	0.3	1.4	n.s.
Delayed Recall (% errors)	0.3	1.3	n.s.
Paired Associate Visual learning			
(% errors)	0.7	1.9	<i>p</i> = .02
Overall (excluding IQ)			
Mean	0.5	0.8	001. > מ

Table 3 P are in bolo	'earson ł. ^a = si	ı r correlâ gnificant	ations between Phe <.05; ^b significant	e measures t <.01. To facil	aken at differ itate interpre	ent points tation, posi	in time and adu tive correlations	ult cognitive s always indi	perforr cate th	nance (at high	N partion Phe wa	cipants = 5 Is associate	1–56; N tas ed with wo	sks = 16 orse per). Significa formance	Int correlations Thus, for IQ,
digit span,	Corsi 5	span, anc	l semantic fluency	correlations	were reverse	q										
PHE			Visual attent	ion speed		Visuo Mot	or coordination		EF //	Aonitori	ng		Sustained		earning ar	nd memory
	FSIQ	Simple	Detection	Feature	Conjunction	Peg-	Digit	WCST	TMT	Digit	Corsi	Semantic	Attention	Rey	Rey	Paired Associate
		RT	with distractors RT	Search RT	Search RT	board	Symbol	Total errors	B-A	span	Span	Fluency	RVP %	a1-a5	delayed	Visual learning
0-10 yrs																
Average	.20	.22	.30 ^a	.33 ^a	.34 ^a	.18	.18	60.	.28 ^a	.12	02	.28	.12	90.	.07	.34 ^a
SD	.43 ^b	.43 ^b	.56 ^b	.41 ^b	.44 ^b	.40 ^b	.15	.19	.47 ^b	.37 ^a	90.	.25	.31 ^a	.35 ^a	.36 ^a	.53 ^b
11-16 yrs																
Average	.32 ^a	.14	.18	.33 ^a	.27	.28 ^a	.27	.32 ^a	.22	.11	60.	.34 ^a	.32 ^a	.17	.14	.60 ^b
SD	.40 ^b	.16	.25	.25	.20	.40 ^b	03	.21	.27	.30 ^a	06	.42 ^b	.32 ^a	.07	E.	.42 ^b
17 yr to nov	Z															
Average	.42 ^b	.05	.29 ^a	.35 ^a	.32 ^a	.24	.35 ^a	.31 ^a	.43 ^b	.10	.05	.21	.38 ⁵	.28	.29 ^a	.44 ^b
SD	.47 ^b	90.	.26	.23	.15	.33 ^a	00.	.26	.54 ^b	.12	-0.1	.27 ^a	.37 ^b	03	90.	.23
Lifetime																
Average	.23	.13	.27	.40 ^b	.40 ^b	.18	0.28	.29 ^a	.28 ^a	.01	.04	.16	91.	.12	.12	.40 ^b
SD	.56 ^b	.23	.43 ^b	.29 ^a	.24	.39 ^b	0.07	.28 ^a	.47 ^b	.26	-0.04	.29 ^a	.40 ^b	.21	.29 ^a	.48 ^b
Current	.43 ^b	00:	0.16	60.	60.	.26	.34 ^a	.22	.55 ^b	.05	-0.11	.17	.42 ^b	.34 ^a	.39 ^b	.33 ^a
Note. Correlé the same val	ations hi lue of r ı	ave been (may have	carried out using z sco different probabilities	res. For the En	glish PKU samp	le the numbe	r of participants a	nd available Ph	ne meast	ures is sli	ghtly dif	erent across	tests (range	31–37 p	articipants),	this means that

correlations between Phe measures taken at different points in time and adult cognitive performance (N participants = 51–56, N tasks = 16). Significant correlation	ificant <.05; ^b significant <.01. To facilitate interpretation, positive correlations always indicate that high Phe was associated with worse performance. Thus, for IC	an, and semantic fluency correlations were reversed
ile 3 Pearson r correlations betw	in bold. ^a = significant <:05; ^b sig	t span, Corsi span, and semantic

semantic fluency), sustained attention, Rey words delayed recall, and paired visual learning.

Consistent with previous results [12], tasks tapping visuo-attentional speed were associated with blood Phe early in life, but less with adult blood Phe and not at all with current Phe level. AwPKU who had maintained a more constant control in early childhood (0–10 years) still showed positive effects many years later, in adulthood, with faster RTs. In contrast, other tasks correlated strongly even with current Phe level. FSIQ, visuo-motor coordination (digit symbol), sustained attention, TMT B-A and learning are all strongly affected by current Phe level (as well as by levels at previous years).

Phe average vs Phe SD

Data analyses

Effects of Phe average and Phe SD were compared with different analyses. We compared the effect of these measures at different ages by contrasting correlations between Phe average/Phe SD in either childhood or adulthood and adult cognitive outcomes. We compared the number of significant correlations through χ^2 tests and average size of correlation with t-tests.

Furthermore, we compared the relative contribution of Phe average and Phe SD to cognition by carrying out regression analyses where cognition was measured with either IQ or mean z-score in our cognitive battery as a summary measure of performance (contribution of individual measures is shown in the previous section with correlation analyses). We carried out three types of regressions. First of all, we compared the effects of Phe average and Phe variation across the lifespan. We carried out a two-steps regression where education was entered in the first step (to partial out any contribution) and both Phe average and Phe variation were entered together in the second step (forward method where the variables making the strongest contribution is considered first and, then, any other variable which makes an additional significant contribution is added). Note that entering education at a first step is a conservative choice, not only because there is a mutual relationship between IQ and education (with education influencing IQ, but also IQ influencing education), but also because Phe levels may influence education. In a second analysis, we assessed directly the contribution of Phe SD after Phe average was considered. Therefore, Phe average was forced in the first step and Phe variation was entered in the second step. Finally, we carried out a third type of regression to consider the contribution of metabolic measures at different ages. Based on the correlation results, we contrasted Phe average and Phe variation taken in childhood with the same measures taken either in adolescence or adulthood. All measures were entered together in the regression equation to see which combination predicted cognition best (SPSS forward method). In this analysis the order in which the variables are entered in the equation is identified by the regression model. The variable making a stronger contribution is entered first following by any other variable making an additional, significant contribution. We considered either adult or adolescent values in separate analyses because of their high correlation (for Phe average r = .74; for Phe variation r = .50) and becaue we wanted to avoid power with more variables.

Results

Relative contribution of Phe average vs Phe SD in childhood vs adulthood

An inspection of Table 3 suggests that different metabolic measures have different effect on cognition at dif-Considering ages further apart, ferent ages. we statistically compared correlations with Phe average and Phe SD in childhood and adulthood. Within childhood measures, there was a higher number of significant correlations and a higher mean correlation with Phe SD than with Phe average (12/16 vs 5/16; χ^2 = 6.1; p = .01; Pearson r = .30 vs .16; t-test = 6.4; p < .001). The opposite was true for adult measures. Here, there was a higher number of significant correlations and a higher mean correlation with Phe average than with Phe SD (10/16 vs 5/16; $\chi^2 = 3.1$; p = .08; Pearson r = .27 vs .19; t-test = 2.3 p < .03). Importantly, the different degree of association of Phe variations with cognitive performance at different points in life was not due to differences in variability since Phe variation was similar across the life span (see Table 1). These results suggest that Phe average and Phe variation impact on the brain through different mechanisms since their effect differ at different ages. If this is the case, we should be able to demonstrate that both these measures contribute independently to explain adult cognitive outcomes.

Independent contribution of Phe average vs Phe SD

Results of regression analyses are shown in Tables 4 and 5. Table 4 shows two analyses. Analysis A is a two-steps regression where education was entered in the first step and Phe average and Phe SD were entered together in the second step. Results suggest that education influences IQ, but not cognitive performance in our battery (mean z-score). Crucially, results also show that Phe SD is the main predictor of both IQ and overall z-score. Phe average, however, also makes a (marginally significant) contribution to explain mean z-score. Analysis B is a two-steps regression where Phe average was entered in the first step and Phe SD in the second step. Results indicate that Phe SD is a significant predictor of performance even when the contribution of Phe average is partialled out, explaining a significant additional amount of

Table 4 Regression analyses predicting cognitive outcomes from Phe variation (SD) and Phe average entered either at the same step (forward method; set A) or at different steps (set b). Phe variation and Phe average are computed across the life-span. Coefficient and p refer to best model

Dependent variable		R2	Adj R2	Predictors	stand β	р	R2 change
Analysis-A							
Full IQ	Step 1	0.16	0.14	Education	.28	.01	
	Step 2	0.39	0.37	Phe SD	-0.5	<.001	
				Phe average	-	-	
Mean z-score	Step 1	0.09	0.08	Education	-19	.09	
	Step 2	0.42	0.39	Phe SD	.46	<.001	
				Phe average	.23	.049	
Analysis B							
Full IQ	Step 1	0.05	0.03	Phe average	03	n.s.	-
	Step 2	0.32	0.29	Phe SD	55	<.001	0.27
Mean z-score	Step 1	0.16	0.14	Phe average	.22	.06	-
	Step 2	0.39	0.36	Phe SD	.51	<.001	0.23

variation both in IQ (27%) and mean z-score (23%). Phe average makes a smaller, marginally significant contribution, explaining 16% of variance in mean z-scores.

Table 5 shows the relative contribution of Phe measures (average and SD) taken either during childhood or adolescence/adulthood. Results show that IQ is predicted only by adolescent/adult Phe average. Mean zscore instead is predicted by *both* childhood Phe SD and adolescent/adult Phe average. These variables together predict a particularly high proportion of variance in mean z-score (41 and 43% respectively in a regression including either adolescent or adult Phe average).

Conclusion

Overall these results highlight the importance of considering both Phe variation (especially in childhood) and Phe average levels (especially in adulthood) as predictors of adult cognitive outcomes.

Individual variability in cognitive outcomes Data analyses

To examine individual variability in cognitive outcomes, we performed three different types of analyses.

First of all, we compared the distribution of cognitive scores in PKU and control participants to see if shifts in performance involved the whole distribution or only the bottom part of the distribution. For this purpose, we have divided both PKU and control groups into thirds according to best vs worse performance for IQ and mean z-score. Then, we have performed a between-subjects Anova with group (PKU vs control) and subgroup (top third vs bottom third) as independent variables and either IQ or mean zscore as a dependent variable.

Second, we examined the distribution of cognitive scores in relation to metabolic control. We first visually

examined scatter plots to see if the Phe boundaries identified by current European guidelines were meaningful in eliminating/reducing individuals with poor scores. Then, we used χ^2 tests to compare the proportions of individuals with poor cognitive scores whose metabolic control was or not within the guideline boundaries. Arbitrarily, we considered 'poor' scores, which were = > 1 z-score from the control group. In a normal distribution, this would apply to 16% of scores.

Lastly, we assessed if we could identify any exceptional cases in our data-base where poor Phe control was associated with good cognitive performance. To be conservative, we identified as having poor control any individuals with average Phe levels (in either childhood and adolescence) exceeding both European guidelines and the average for the PKU group (= > 500 in childhood and = > 900 in either adolescence and adulthood). To be conservative, we identified as

Table 5 Simultaneous regression analyses –method forward– predicting cognitive outcomes from: a) Childhood Phe average; b) childhood Phe SD; c) adolescent/adult Phe average; d) adolescent/adult Phe SD. Analysis A uses adolescent measures; Analysis B uses adult measures

Analysis D uses	addit measures				
Dependent	Predictors	Best	: model		
Variables	Entered	R2	Adj R2	stand β	р
Analysis - A					
Full IQ	Adolescent Phe average	.23	.21	-0.48	.001
Mean z-score Adolescent Phe average Childhood Phe SD		.43	.41	0.38	.01
				0.36	.01
Analysis - B					
Full IQ	Adult Phe average	.22	.20	-0.47	.001
Mean z-score	Childhood Phe SD	.41	.39	0.41	.005
	Adult Phe average			0.32	.02

having good cognition any individuals where IQ was within .05 z-score from controls, mean z-score was average (=0) or better than average (negative), and where there was a normal incidence of very poor z-scores (= >1.5) across tasks (this proportion is on average 6% in the control group).

Results

Distribution of cognitive scores in PKU and control groups

Figure 1 compares results for the PKU and control subgroups depending on how their performance ranked within their group (best or worse third). Results were mixed. The PKU group was always worse than the control group, but differences were more marked for the worst subgroups. In fact, there was a significant interaction of group (PKU vs control) x subgroup (best vs worst third) with z-scores (F (1,65) =7.1; p = .01) and a marginal difference with IQ (F (1, 65) =2.7; p = .10). These results indicate that, if there is a fixed cost of having PKU, this is marginal.

Distribution of scores in relation to metabolic control

Figure 2 shows the distributions of IQ scores and z-scores in relation to metabolic measures: Phe average and Phe variation in childhood (Panel A) and Phe average and Phe variation in adolescence/adulthood (Panel B). To reduce the number of plots, we have averaged measures in adolescence and adulthood since these are highly correlated and similarly related to cognitive outcomes as shown by the regression analyses. With few exceptions, poor scores correspond to individuals who have not followed the guidelines (Phe levels = > 360 in childhood and/or = > 600 after that). Additionally, there are few or no individuals with poor zscores and Phe variation < 180 in childhood and in adolescence/adulthood. There are more individuals with poor IQ and Phe variation < 180 in adolescence/adulthood, but note that even here Phe variation is a good predictor of performance as shown by the slope of the regression line.

Table 6 shows the number and % of individuals with poor cognitive scores in relation to guideline adherence. Numbers are small so exact values are not very meaningful, but patterns are clear. In all cases, there is a higher % of individuals with poor scores among those who have not followed guidelines and in most cases differences are significant. In particular, the risk of poor cognitive performance is about 30% higher in individuals who have relaxed the diet after childhood.³

Exceptional cases

In our sample, following our definition, there were 13 participants with poor metabolic control in childhood

(average Phe= > 500); 12 with poor metabolic control in adolescence (average Phe = > 900) and 27 with poor metabolic control in adulthood (average Phe= > 900). Following our criteria, out of these participants, we could identify only 3 individuals with completely normal cognition and poor metabolic control in adulthood (3/ 27 = 11%). Their profile is shown in Table 7. We could not identify any individuals with poor control in childhood and adolescence and good cognition, but our samples are small. Also note that our definition of 'good' cognition was strict and included performance in our ad-hoc PKU cognitive battery. If we consider cognition only in terms of IQ, a larger proportion of participants have allegedly normal cognition (IQ within .5 SD from the control mean), but poor control in childhood: 4/13; adolescence: 4/12; and adulthood: 6/27. Across ages, we found 14/52 exception cases considering only IQ vs 3/54 considering cognition more widely ($\chi^2 = 7.0$; p = .008).

Discussion

Our study had two main aims: 1. to compare the effects of average Phe levels and Phe variation/fluctuation (in terms of Phe SD) on cognitive functions and 2. to explore cognitive variability in relation to metabolic control in a population of adults with PKU.

First of all, our results showed developmental interactions between type of cognitive function and type of metabolic measure (average vs SD). Different functions were affected by historical vs current metabolic control and by Phe average vs. Phe SD. Speed of processing in visual search was affected by metabolic control in childhood measured by both Phe median levels and Phe SD, while current Phe level had little impact [57]. Other tasks correlated significantly with current Phe --especially those involving visuo-motor coordination (digit symbol), learning and memory, an executive component (reasoning: IQ; flexibility: TMT B-A) and sustained attention (RVP). Consistent with our results, Moyle et al. [18] also found that a group of young adults with PKU (N = 12) showed significant differences from controls (N = 12) with the Processing Speed Index from the WAIS, but no correlations with *current* Phe level. The opposite was true for a memory index (WMS-III), where there was no impairment, but a strong correlation with *current* Phe levels (except for short-term memory, as in our case). Additionally, our regression analyses showed that, overall, adult cognition was best predicted by a combination of Phe childhood SD and Phe adolescence/adult averages.

These results suggest that there are different mechanisms through which Phe impact on cognition. High Phe may cause changes in levels of neurotransmitters (e.g., lowering dopamine) which can be modulated relatively short-term [60]. Alternatively, high Phe may affect brain structures (e.g., white matter; see Anderson et al. [61];

³Results were in the right direction but did not reach significance when we considered childhood average levels. This may simply be error variation due to our small sample; childhood Phe average, in fact, was also significantly correlated with adult outcomes.



Anderson & Leuzzi, [62]) in a way which can be effective only during some critical periods and/or can be appreciated only long-term. Our results do not support one hypothesis of impairment versus another, but indicate that different mechanisms must be at play. This is necessary to explain our results showing that Phe average and Phe SD have a different impact on cognition at different ages. Possibly, Phe peaks are more detrimental for developing brains because they impact on white matter structural integrity [18, 63], while cumulative effects of Phe --as indexed by Phe averages—may be more detrimental for adult brains because they also impact on levels of neurotransmitters.

Clinically, our results indicate the importance of keeping low both Phe average and Phe variation. Regression analyses showed that Phe variation throughout the lifespan was a stronger predictor of IQ and performance in our cognitive battery (mean z-score) although both Phe average and Phe variation across the life-span contributed to determine mean z-score. Together, childhood Phe variation and adult average levels accounted for as much as 41% of variability in overall z-score.

Regarding cognitive variability, our results confirmed extreme variability in cognitive outcomes across individual PKU participants. AwPKU in the top third of a distribution of cognitive scores performed very well. In fact, they performed almost as well as the control participants in the top third of their distribution. This suggests that the cognitive costs of PKU are minimal beyond those associated with high levels of Phe. If Phe levels are kept in check, performance can be (almost) as good as in the best controls. Minor costs could be associated to living with a chronic disease and/or with the fact that Phe control is not perfect even in individuals with top cognitive performance. This is in contrast with participants in the bottom third of the distribution who showed substantial impairments across a wide range of tasks.

Importantly, considering the distributions of cognitive scores in relation to metabolic control, we found that maintaining Phe < 600 after childhood significantly

reduced the risk of poor cognitive outcomes by around 30%. This was true both when performance was measured by IQ and when it was measured as mean z-score in our cognitive battery. We also found that reduced Phe variation in childhood and afterwards significantly reduced the risk of poor outcomes (= > 1 z-score from controls) by a similar amount. These results, together with the results of our regression analyses, provide evidence for maintaining a good *and stable* dietary control after childhood. Distinguishing the contribution of metabolic control in adulthood and adolescence is more difficult given that these measures are highly intercorrelated. However, the significant effect of current Phe on specific tasks suggests that dietary control continuously modulates cognition, at least in some domains.

Our sample included a few individuals with poor metabolic control in adulthood, but excellent cognition. It is difficult to know if these individuals could have achieved even better cognition if they had maintained a low Phe diet, but these cases do raise the question of some possible protective factors in some individuals where the effects of high Phe levels are not evident. Reduced transport of Phe through the blood-brain-barrier could be one such factor although high levels of Phe were still found in the CSF of one these individuals described in the literature [64]. Recently van Vliet et al. [37] has reviewed exceptional cases from the literature of untreated individuals with PKU who have escaped mental disability. One has to stress, however, that these cases are very few. In our sample there were only three cases with good cognition and poor metabolic control in adulthood (Phe $= > 900 \,\mu/L; N = 3/27 = 11\%$) and no cases with good cognition and poor metabolic control at earlier ages.

Finally, we note the importance of using ad-hoc, comprehensive batteries to test and monitor cognition in people with PKU. Phe measures were more related to variance in our cognitive battery than in IQ (see Table 5). Moreover, the proportion of individuals who apparently achieved good



cognition in spite of poor metabolic control was higher when cognition was measured only through IQ. It is important to consider cognition comprehensively to avoid false negatives of no consequences of relaxing the diet when, in fact, there is a clear impact in some cognitive areas.

Study limitations

The main limitation of our study is the number of participants which interacts with the characteristics of metabolic control in samples of adults with PKU. It is difficult to evaluate whether metabolic control in adulthood contributes to cognitive outcomes above metabolic control in adolescence since these two measures are highly correlated. We need larger samples where these variables dissociate. Moreover, metabolic control is often good in childhood and adolescence and this reduces sample sizes when we want to assess the consequences of following the diet less strictly at these ages. Again, a larger sample will help to identify these cases. More

Table 6 Number and percentages of poor cognitive scores in individuals who have followed or not followed European guidelines (Phe average < 360 in childhood; < 600 after that); Poor cognitive scores = > 1 SD worse than the controls. Z-score = mean z score in our cognitive battery

			Childhood 0-	-10			Adolescence/Adu	lt 11+	
		Guid	elines	Diff.	χ2	Guid	elines	Diff.	χ2
		~	S			~	8		
Phe AVERAG	E	Phe < 360	Phe > 360		р	Phe < 600	Phe > 600		р
IQ	Ν	10/22	16/28			4/15	23/41		
	%	45.5	57.1	11.7	.30	26.7	56.1	29.4	.05
z-score	Ν	3/22	9/28			0/15	12/41		
	%	13.6	26.5	12.8	.12	0	29.3	29.3	.01
Phe SD		Phe < 180	Phe > 180			Phe < 180	Phe > 180		
IQ	Ν	5/14	21/34			15/38	11/18		
	%	35.7	61.8	26.1	.09	42.2	61.1	19.0	.11
z-score	Ν	0/14	12/34			5/38	-7/18		
	%	0	35.3	35.3	.008	13.6	38.9	25.7	.03

Table 7 Performance of potentially exceptional participants with good cognition in spite of poor metabolic control or vice-versa poor cognition in spite of good control (see text for further explanation). Childhood Phe results for PKU108 are not available

	Good cognition but poor adult control				
	PKU4	PKU107	PKU108		
	IT	UK	UK		
Age	24	40	27		
Education (in years)	13	14	16		
Gender (M/F)	Μ	Μ	Μ		
Full IQ	109	117	126		
Feature search RT	-0.1	-0.3	-1.1		
Conjoined search RT	0.1	-0.5	-1.4		
Pegboard	0.1	-0.6	1.1		
Average Z score	0.1	0.0	-0.2		
% z score = > 1.5	4.2	8.3	8.3		
Childhood (0–10 y)					
Phe Average Median	498	425	-		
Phe Variation	181	213	-		
Adolescence (11–16 y)					
Phe Average Median	634	622	845		
Phe Variation	634	113	150		
Adulthood (17 y +)					
Phe Average Median	1067	1115	1000		
Phe Variation	239	114	137		
Lifetime					
Phe Average Median	672	978	975		
Phe Variation	169	148	139		
Current Phe	1635	1080	685		

collaborations across clinics are necessary to achieve adequate sample sizes.

Conclusions

Our study has furthered our understanding of PKU in two main ways. We have demonstrated developmental effects where outcomes depend on complex interactions between the cognitive function examined, the age when metabolic control is measured (childhood, adolescence, current level) and the type of metabolic variable considered (Phe average vs Phe variation). Significant effect of childhood Phe levels on adult cognitive performance decades later support the idea of critical periods for brain development where high Phe levels and particularly Phe peaks (Phe variation) are very detrimental [65]. Significant correlations between *current* or *adult* average Phe levels and cognitive performance are consistent with the alternative, but not mutually exclusive, view that the toxic effect of Phe cumulate throughout the lifetime (for an overview on this topic see Berry et al. [66]).

Secondly, our study has provided evidence broadly consistent with the safety of following current European guidelines to avoid cognitive impairments. Levels < 360 μ mo/l in childhood and < 600 μ mo/l in adulthood prevent cognitive impairments. We have also suggested, however, that maintaining low Phe variation should be an equally important criterion when setting guidelines. In, our sample SD < 180 was also instrumental in preventing poor cognitive scores. While guidelines, should be followed to avoid poor outcomes, the presence of a few cases with high adult Phe level, but excellent cognition suggests that there is individual variability in Phe sensitivity. Given the social and economic costs of maintaining a low Phe diet, identifying who and why may avoid a strict diet without cognitive drawbacks should be one of the priorities of future research.

Appendix

Table 8 Raw scores for PKU (N = 50-56) and controls (N = 49); Highlighted results are significantly different between the two groups (see Table 2)

	PKU raw score		Controls raw score	
	Mean	SD	Mean	SD
Visual attention RTs				
Simple Detection - ms.	329.3	49.9	315.5	52.0
Detention with Distractors – ms.	464.2	93.0	411.8	63.8
Feature Search – ms.	661.5	217.9	532.2	98.5
Conjunction Search m.	1062.7	246.6	878.3	165.4
Visual attention accuracy				
Detention with Distractors - % errors	0.8	1.0	0.6	0.9
Feature Search - % errors	1.6	3.5	1.5	2.1
Conjunction Search - % errors	2.2	3.5	1.6	4.5
Visuo-motor coordination				
Pegboard (sec.)	73.6	13.1	67.6	7.9
Digit Symbol (%errors in 90 s.)	38.9	11.7	26.0	10.0
Trail-Making Test A: (sec.)	27.6	12.0	23.8	9.6
Executive functions				
WCST				
Total errors	13.9	7.7	11.9	5.2
Perseverative responses	7.7	4.5	7.2	4.0
N of Completed Categories	3.9	1.1	4.3	0.9
Trail-Making Tests				
B (sec.)	56.2	29.2	45.9	17.5
B-A (sec.)	28.7	23.2	22.1	13.2
Verbal Fluency				
Letter (correct answers)	37.0	11.1	44.0	11.9
Semantic (correct answers)	20.9	5.7	24.4	4.52
Short term memory				
Digit span	6.0	1.0	6.3	1.0
Corsi Block span	5.5	0.9	5.4	1.0
Sustained attention				
RVP (% of errors)	20.6	11.8	14.3	9.1
Learning				
Rey Auditory Verbal Learning Test				
Trial A1-A5 (% errors)	22.8	10.4	20.2	7.5
Retention (% errors A6)	17.9	14.5	12.9	13.8
Delayed Recall (% errors)	14.0	14.3	11.2	13.4
Paired Associate Visual learning				
(% errors)	2.8	3.3	1.7	1.7

Abbreviations

AwPKU: Adults with PKU; EF: Executive functions; Phe: Phenylalanine; PKU: Phenylketonuria; RVP: Rapid visual processing; SD: Standard Deviation; STM: Short-term memory; TMT: Trial making test; WCST: Wisconsin card sorting test; µmol/L: Micromole per liter

Acknowledgements

We would like to thank Tarekegn Gerberhiwot for having allowed access to the PKU participants, Cecilia Guariglia for having facilitated tested of the Italian control participants and Andrew Olson for helpful comments.

Authors' contributions

CR, LP, AMC, VL: Study conception and design, data analysis and interpretation of data; CR also drafting and revision of manuscript. FM, FN supervision of data collection, contribution to critical interpretation and critical revision of manuscript. FV, NF, CC: Data collection and analyses. All authors reviewed early and final drafts of the manuscript and were fully responsible for the content and editorial decisions related to this manuscript. All authors read and approved the final manuscript.

Funding

The original cohort of English patients was supported by a Marie Curie Intra European Fellowship within the 7th European Community Framework Programme granted to Liana Palermo under the supervision of Cristina Romani and by a grant of the University Hospital Birmingham Charity to Tarekegn Gerberhiwot.

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

The English study received NHS ethical approval. The Italian study was approved by the local ethics committee. All participants provided informed consent to the study.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹School of Life and Health Sciences, Aston University, Aston Triangle, Birmingham, England B4 7ET, UK. ²Department of Human Neuroscience – Unit of Child Neurology and Psychiatry, Sapienza University of Rome, Rome, Italy. ³Department of Psychology, Sapienza University of Rome, Rome, Italy. ⁴Department of Experimental Medicine, Sapienza University of Rome, Rome, Italy. ⁵Department of Clinical Medicine, Policlinico Umberto I, Rome, Italy. ⁶Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK. ⁷Department of Medical and Surgical Sciences, Magna Graecia University of Catanzaro, Catanzaro, Italy.

Received: 19 June 2019 Accepted: 9 October 2019 Published online: 28 November 2019

References

- de Groot MJ, Hoeksma M, Blau N, Reijngoud DJ, van Spronsen FJ. Pathogenesis of cognitive dysfunction in phenylketonuria: review of hypotheses. Mol Genet Metab. 2010;99(supp 1):S86–9. https://doi.org/10. 1016/j.ymgme.2009.10.016 Review.
- Waisbren SE, Noel K, Fahrbach K, Cella C, Frame D, Dorenbaum A, Levy H. Phenylalanine blood levels and clinical outcomes in phenylketonuria: a systematic literature review and meta-analysis. Mol Genet Metab. 2007; 92(1–2):63–70. https://doi.org/10.1016/j.ymgme.2007.05.006.
- Hofman DL, Champ CL, Lawton CL, Henderson M, Dye L. A systematic review of cognitive functioning in early treated adults with phenylketonuria. Orphanet J Rare Dis. 2018;13(1):150. https://doi.org/10.1186/s13023-018-0893-4.
- Leuzzi V, Tosetti M, Montanaro D, Carducci C, Artiola C, Carducci C, et al. The pathogenesis of the white matter abnormalities in phenylketonuria. A

- Pietz J, Kreis R, Boesch C, Penzien J, Rating D, Herschkowitz N. The dynamics of brain concentrations of phenylalanine and its clinical significance in patients with phenylketonuria determined by in vivo 1H magnetic resonance spectroscopy. Pediatr Res. 1995;38:657–63. https://doi.org/10. 1203/00006450-199511000-00005.
- Rupp A, Kreis R, Zschocke J, Slotboom J, Boesch C, Rating D, Pietz J. Variability of blood—brain ratios of phenylalanine in typical patients with phenylketonuria. J Cereb Blood Flow Metab. 2001;21:276–84. https://doi.org/ 10.1097/00004647-200103000-00011.
- Brumm VL, Azen C, Moats RA, Stern AM, Broomand C, Nelson MD. Neuropsychological outcome of subjects participating in the PKU adult collaborative study: A preliminary review. J Inherit Metab Dis. 2004;27(5): 549–66. https://doi.org/10.1023/B:BOLI.0000042985.02049.ff.
- Moats RA, Koch R, Moseley K, Guldberg P, Güttler F, Boles RG, Nelson MD. Brain phenylalanine concentration in the management of adults with phenylketonuria. J Inherit Metab Dis. 2000;23:7–14.
- Schindeler S, Ghosh-Jerath S, Thompson S, Rocca A, Joy P, Kemp A, Rae C, et al. The effects of large neutral amino acid supplements in PKU: an MRS and neuropsychological study. Mol Genet Metab. 2007;91:48–54. https://doi. org/10.1016/j.ymgme.2007.02.002.
- Pietz J, Dunckelmann R, Rupp A, Rating D, Meinck HM, Schmidt H, et al. Neurological outcome in adult patients with early-treated phenylketonuria. Eur J Pediatr. 1998;157(10):824–30. https://doi.org/10.1007/s004310050945.
- Vilaseca MA, Lambruschini N, Gómez-López L, Gutiérrez A, Fusté E, Gassió R, et al. Quality of dietary control in phenylketonuric patients and its relationship with general intelligence. Nutr Hosp. 2010;25(1):60–6.
- Romani C, Palermo L, MacDonald A, Limback E, Hall SK, Geberhiwot T. The impact of phenylalanine levels on cognitive outcomes in adult patients with phenylketonuria: effects across tasks and developmental stage. Neuropsychology. 2017;31(3):242–54. https://doi.org/10.1037/neu0000336.
- Manti F, Nardecchia F, Paci S, Chiarotti F, Carducci C, Carducci C, et al. Predictability and inconsistencies in the cognitive outcome of early treated PKU patients. J Inherit Metab Dis. 2017;40(6):793–9. https://doi.org/10.1007/ s10545-017-0082-y.
- Weglage J, Fromm J, van Teeffelen-Heithoff A, Möller HE, Koletzko B, Marquardt T, et al. Neurocognitive functioning in adults with phenylketonuria: results of a long-term study. Mol Genet Metab. 2013; 110(Suppl):S44–8. https://doi.org/10.1016/j.ymgme.2013.08.013.
- Bartus A, Palasti F, Juhasz E, Kiss E, Simonova E, Sumanszki C, et al. The influence of blood phenylalanine levels on neurocognitive function in adult PKU patients. Metab Brain Dis. 2018;33(5):1609–15. https://doi.org/10.1007/ s11011-018-0267-6 Epub 2018 Jun 12.
- Channon S, Goodman G, Zlotowitz S, Mockler C, Lee PJ. Effects of dietary management of phenylketonuria on long-term cognitive outcome. Arch Dis Child. 2007;92(3):213–8. https://doi.org/10.1136/adc.2006.104786.
- Jahja R, Huijbregts SCJ, de Sonneville LMJ, van der Meere JJ, Legemaat AM, Bosch AM, et al. Cognitive profile and mental health in adult phenylketonuria: a PKU-COBESO study. Neuropsychology. 2017;31(4):437–47. https://doi.org/10.1037/neu0000358.
- Moyle JJ, Fox AM, Bynevelt M, Arthur M, Burnett JR. A neuropsychological profile of off-diet adults with phenylketonuria. J Clin Exp Neuropsychol. 2007;29(4):436–41. https://doi.org/10.1080/13803390600745829.
- Koch R, Burton B, Hoganson G, Peterson R, Rhead W, Rouse B, et al. Phenylketonuria in adulthood: a collaborative study. J Inherit Metab Dis. 2002;25(5):333–46.
- Feldmann R, Denecke J, Grenzebach M, Weglage J. Frontal lobe-dependent functions in treated phenylketonuria: blood phenylalanine concentrations and long-term deficits in adolescents young adults. J Inherit Metab Dis. 2005;28(4):445–55. https://doi.org/10.1007/s10545-005-0445-7.
- Anastasoaie V, Kurzius L, Forbes P, Waisbren S. Stability of blood phenylalanine levels and IQ in children with phenylketonuria. Mol Genet Metab. 2008;95(1–2):17–20. https://doi.org/10.1016/ j.ymgme. 2008.06.014.
- 22. Arnold GL, Kramer BM, Kirby RS, Plumeau PB, Blakely EM, Sanger Cregan LS, et al. Factors affecting cognitive, motor, behavioral and executive functioning in children with phenylketonuria. Acta Paediatr. 1998;87(5):565–70.
- Viau KS, Wengreen HJ, Ernst SL, Cantor NL, Furtado LV, Longo N. Correlation of age-specific phenylalanine levels with intellectual outcome in patients with phenylketonuria. J Inherit Metab Dis. 2011;34(4):963–71. https://doi.org/ 10.1007/s10545-011-9329-1.

- Burgard P, Schmidt E, Rupp A, Schneider W, Bremer HJ. Intellectual development of the patients of the German Collaborative Study of children treated for phenylketonuria. Eur J Pediatr. 1996;155(supp): S33–8.
- Hood A, Grange DK, Christ SE, Steiner R, White DA. Variability in phenylalanine control predicts IQ and executive abilities in children with phenylketonuria. Mol Genet Metab. 2014;111(4):445–51. https://doi.org/10. 1016/j.ymgme.2014.01.012.
- Jahja R, Huijbregts SC, de Sonneville LM, van der Meere JJ, van Spronsen FJ. Neurocognitive evidence for revision of treatment targets and guidelines for phenylketonuria. J Pediatr. 2014;164(4):895–9. https://doi.org/10.1016/j. jpeds.2013.12.015 Epub 2014 Jan 30.
- Hood A, Antenor-Dorsey JA, Rutlin J, Hershey T, Shimony JS, McKinstry RC, et al. Prolonged exposure to high and variable phenylalanine levels over the lifetime predicts brain white matter integrity in children with phenylketonuria. Mol Genet Metab. 2015;114(1):19–24. https://doi.org/10. 1016/j.ymgme.2014.11.007.
- Cleary M, Trefz F, Muntau AC, Feillet F, van Spronsen FJ, Burlina A. Fluctuations in phenylalanine concentrations in phenylketonuria: a review of possible relationships with outcomes. Mol Genet Metab. 2013;110(4):418–23. https://doi.org/10.1016/j.ymgme.2013.09.001.
- van Spronsen FJ, van Wegberg AM, Ahring K, Bélanger-Quintana A, Blau N, Bosch AM, et al. Key European guidelines for the diagnosis and management of patients with phenylketonuria. Lancet Diabetes Endocrinol. 2017;5(9):743–56. https://doi.org/10.1016/S2213-8587(16)30320-5.
- van Wegberg AMJ, MacDonald A, Ahring K, Bélanger-Quintana A, Blau N, Bosch AM, et al. The complete European guidelines on phenylketonuria: diagnosis and treatment. Orphanet J Rare Dis. 2017;12(1):162. https://doi. org/10.1186/s13023-017-0685-2 Review.
- Vockley J, Andersson HC, Antshel KM, Braverman NE, Burton BK, Frazier DM, et al. Phenylalanine hydroxylase deficiency: diagnosis and management guideline. Genet Med. 2014;16(2):188–200. https://doi.org/10.1038/ gim.2013.157.
- Burgard P, Ullrich K, Ballhausen D, Hennermann JB, Hollak CEM, Langeveld M, et al. Issues with European guidelines for phenylketonuria. Lancet Diabetes Endocrinol. 2017;5(9):681–3. https://doi.org/10.1016/ S2213-8587(17)30201-2.
- Diamond A, Prevor MB, Callender G, Druin DP. Prefrontal cortex cognitive deficits in children treated early and continuously for PKU. Monogr Soc Res Child Dev. 1997;62(4):1–208.
- Leuzzi V, Carducci C, Carducci C, Chiarotti F, Artiola C, Giovanniello T, Antonozzi I. The spectrum of phenylalanine variations under tetrahydrobiopterin load in subjects affected by phenylalanine hydroxylase deficiency. J Inherit Metab Dis. 2006;29(1):38–46.
- Fonnesbeck CJ, McPheeters ML, Krishnaswami S, Lindegren ML, Reimschisel T. Estimating the probability of IQ impairment from blood phenylalanine for phenylketonuria patients: a hierarchical meta-analysis. J Inherit Metab Dis. 2013; 36(5):757–66. https://doi.org/10.1007/s10545-012-9564-0 Epub 2012 Nov 30.
- Weglage J, Pietsch M, Feldmann R, Koch HG, Zschocke J, Hoffmann G, et al. Normal clinical outcome in untreated subjects with mild hyperphenylalaninemia. Pediatr Res. 2001;49(4):532–6.
- van Vliet D, van Wegberg AMJ, Ahring K, Bik-Multanowski M, Blau N, Bulut FD, et al. Can untreated PKU patients escape from intellectual disability? A systematic review. Orphanet J Rare Dis. 2018;13(1):149. https://doi.org/10. 1186/s13023-018-0890-7 Review.
- Channon S, Mockler C, Lee P. Executive functioning and speed of processing in phenylketonuria. Neuropsychology. 2005;19(5):679–86. https:// doi.org/10.1037/0894-4105.19.5.679.
- Smith ML, Klim P, Mallozzi E, Hanley WB. A test of frontal-specificity hypothesis in the cognitive performance of adults with phenylketonuria. Dev Neuropsychol. 1996;12(3):327–41. https://doi.org/10.1080/ 87565649609540656.
- Palermo L, Gerhiwot T, MacDonald A, Limback E, Hall SK, Romani C. Cognitive outcomes in early-treated adults with phenylketonuria (PKU): a comprehensive picture across domains. Neuropsychology. 2017;31(3):255– 67. https://doi.org/10.1037/neu0000337.
- De Felice S, Romani C, Geberhiwot T, MacDonald A, Palermo L. Language processing and executive functions in early treated adults with phenylketonuria (PKU). Cogn Neuropsychol. 2018;35(3–4):148–70. https://doi. org/10.1080/02643294.2017.1422709.
- 42. Wechsler D. Wechsler adult intelligence scale- revised (WAIS-R). New York: The Psychological Corporation; 1981. Italian version, Laicardi C, Orsini A.

Scala di intelligenza Wechsler per adulti, Firenze: Organizzazioni Speciali; 1997

- 43. Wechsler D. Wechsler abbreviated scale of intelligence (WASI). San Antonio: Harcourt Assessment; 1999.
- 44. Trites R. Grooved pegboard test. Lafayette: Lafayette Instrument; 1977.
- AITB. Army Individual Test Battery. Manual of directions and scoring. Washington, DC: War Department, Adjutant General's Office; 1944.
- Sánchez-Cubillo I, Periáñez JA, Adrover-Roig D, Rodríguez-Sánchez JM, Ríos-Lago M, Tirapu J, Barceló F. Construct validity of the trail making test: role of task-switching, working memory, inhibition/interference control, and visuomotor abilities. J Int Neuropsychol Soc. 2009;15(3):438–50. https://doi. org/10.1017/S1355617709090626.
- Kongs SK, Thompson LL, Iverson GL, Heaton RK. WCST-64: Wisconsin card sorting Test-64 card version, professional manual. Odessa: Psychological Assessment Resources; 2000.
- Novelli G, Papagno C, Capitani E, Laiacona M, Vallar G, Cappa SF. Tre test clinici di ricerca e produzione lessicale. Taratura su soggetti normali. Arch Neurol Psychiatry. 1986;47:477–506.
- Benton AL, Hamsher K d S, Sivan AB. Multilingual aphasia examination. Iowa City: AJA Associates; 1994.
- Rosen WG. Verbal fluency in aging and dementia. J Clin Neuropsychol. 1980;2:135–46.
- Costa A, Bagoj E, Monaco M, Zabberoni S, De Rosa S, Papantonio AM, Carlesimo GA. Standardization and normative data obtained in the Italian population for a new verbal fluency instrument, the phonemic/semantic alternate fluency test. Neurol Sci. 2014;35(3):365–72. https://doi.org/10.1007/ s10072-013-1520-8.
- Corsi PM. Human memory and the medial temporal region of the brain. Montreal: Unpublished doctoral dissertation: McGill University; 1972.
- Sahakian BJ, Jones GMM, Levy R, Gray JA, Warburton DM. The effects of nicotine on attention, information processing, and short-term memory in patients with dementia of the Alzheimer type. Br J Psychiatry. 1989;154:797–800.
- 54. Rey A. L'examen clinique en psychologie [clinical examination in psychology]. Paris: Presses Universitaires de France; 1964.
- Schmidt E, Rupp A, Burgard P, Pietz J, Weglage J, de Sonneville L. Sustained attention in adult phenylketonuria: the influence of the concurrent phenylalanine-blood-level. J Clin Exp Neuropsychol. 1994;16(5):681–8. https://doi.org/10.1080/01688639408402681.
- Sahakian BJ, Morris RG, Evenden JL, Heald A, Levy R, Philpot M, et al. A comparative study of Visuospatial memory and learning in Alzheimer-type dementia and Parkinson's disease. Brain. 1988;111(pt 3):695–718.
- Albrecht J, Garbade SF, Burgard P. Neuropsychological speed tests and blood phenylalanine levels in patients with phenylketonuria: a metaanalysis. Neurosci Biobehav Rev. 2009;33(3):414–21. https://doi.org/10.1016/j. neubiorev.2008.11.001.
- Griffiths P, Paterson L, Harvie A. Neuropsychological effects of subsequent exposure to phenylalanine in adolescents and young adults with earlytreated phenylketonuria. J Intellect Disabil Res. 1995;39(pt 5):365–72. https:// doi.org/10.1111/j.1365-2788.1995.tb00540.x.
- Bik-Multanowski M, Pietrzyk JJ, Mozrzymas R. Routine use of CANTAB system for detection of neuropsychological deficits in patients with PKU. Mol Genet Metab. 2011;102(2):210–3. https://doi.org/10.1016/j. ymgme.2010.10.003.
- Boot E, Hollak CEM, Huijbregts SCJ, Jahja R, Van Vliet D, Nederveen AJ, et al. M. Cerebral dopamine deficiency, plasma monoamine alterations and neurocognitive deficits in adults with phenylketonuria. Psychol Med. 2017; 47(16):2854–65.
- Anderson PJ, Wood SJ, Francis DE, Coleman L, Anderson V, Boneh A. Are neuropsychological impairments in children with early-treated phenylketonuria (PKU) related to white matter abnormalities or elevated phenylalanine levels? Dev Neuropsychol. 2007;32(2):645–68. https://doi.org/ 10.1080/87565640701375963.
- Anderson PJ, Leuzzi V. White matter pathology in phenylketonuria. Mol Genet Metab. 2010;99(suppl 1):S3–9. https://doi.org/10.1016/j.ymgme.2009.10.005.
- Joseph B, Dyer CA. Relationship between myelin production and dopamine synthesis in the PKU mouse brain. J Neurochem. 2003;86(3):615–26. https:// doi.org/10.1046/j.1471-4159.2003.01887.x.
- Allen RJ, Gibson RM. Phenylketonuria with normal intelligence. Am J Dis Child. 1961;102:115–22.

- Abadie V, Rey F, Plainguet F, Rey J. Intellectual development after relaxing the diet at the age of 5 years in typical phenylketonuria. Arch Fr Pediatr. 1992;49(9):773–8.
- Berry SA, Brown C, Grant M, Greene CL, Jurecki E, Koch J, et al. Newborn screening 50 years later: access issues faced by adults with PKU. Genet Med. 2013;15(8):591–9. https://doi.org/10.1038/gim.2013.10.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

